Author’s response to reviews

Title: Gender differences in presentation and diagnosis of chest pain in primary care

Authors:
Stefan Bösner (boesner@staff.uni-marburg.de)
Jörg Haasenritter (haasenri@staff.uni-marburg.de)
Maren A Hani (MarenAbuHani@googlemail.com)
Heidi Keller (kellerhe@staff.uni-marburg.de)
Andreas C Sönnichsen (andreas.soennichsen@pmu.ac.at)
Konstantinos Karatolios (karatoli@med.uni-marburg.de)
Juergen R Schaefer (juergen.schaefer@staff.uni-marburg.de)
Erika Baum (064092007-0001@t-online.de)
Norbert Donner-Banzhoff (norbert@mailer.uni-marburg.de)

Version: 5 Date: 2 December 2009

Author’s response to reviews:

Dear Sir,

I only added on page 5 the explanation for CRF (case report form) and inserted
then on page 7 tow times CRF instead of case report form.

No other changes were made in the manuscript or in the figure and tables.

With kind regards

Stefan Bösner